Author’s response to reviews

Title: Assessment of the effect of application of an educational wiki in flipped classroom on students' achievement and satisfaction

Authors:

Ali Ahmadian Khoynaroud (ahmadian948@gmail.com)
Amirhossein Akbarzadeh (amirhossein_bio@yahoo.com)
Morteza Ghojazadeh (ghojazadehm@hotmail.com)
Saeideh Ghaffarifar (sa.ghafarifar@gmail.com;ghaffarifars@tbzmed.ac.ir)

Version: 2 Date: 25 Aug 2020

Author’s response to reviews:

Dear Editor,

In response to your email on August 25, 2020, I would like to submit the revised version of our manuscript entitled: “Assessment of the effect of application of an educational wiki in flipped classroom on students' achievement and satisfaction”. Really appreciate very invaluable comments from our dear reviewers and their very precise attention. All comments were considered and the requested points were corrected point to point. Amendments are highlighted in yellow and are described as below (divided by their line and page numbers):

Reviewer reports:

Reviewer 2:

• Abstract: please replace the word "evidence" in the conclusion, we don't provide evidence by just one single study, the evidence based process is another complex concept.
  Many thanks for precise attention and your invaluable comment. The amendment was done on page 3, line 57 as: “The findings of this study confirm that a wiki-authoring group assignment is effective in achieving student learning outcomes and …”

• Introduction: I think the hypothesis in line 104-106 needs to be rephrased, you are studying the effect of wiki on FC, so it should be "that using wiki on a FC will promote..." you are not studying the effect of FC because it's applied in both groups.
  Really appreciate your comment and suggestion. The hypothesis was revised as “It is hypothesized that using wiki on a FC will promote students’ attitudes…” on page 5, lines 104-105.

• Methods: line 109, don't repeat "action research" again, and just link the 2 sentences with a connective.
  Thanks a lot for your comment. The revision was done on page 5, line 109 as “This study was an action research, in which, researchers and participants collaboratively work through an iterative process to define a problem …”
line 376, the sentence "A wiki cannot be used to perform any GA in any course" is confusing so please clarify.
Really sorry for that. It was clarified on page 16, line 375 as “A wiki cannot be used to perform any type of GA and in any type of course”. Hope, it is clear this time.